Author's response to reviews

Title: Genetic variation of Glucose Transporter-1 (GLUT1) and albuminuria in 10,278 European Americans and African Americans: a case-control study in the Atherosclerosis Risk in Communities (ARIC) Study

Authors:

Charles C Hsu (cchsu@jhmi.edu)
W.H. L Kao (wkao@jhsphs.edu)
Michael W Steffes (stef001@tc.umn.edu)
Tejal Gambir (trami@jhsphs.edu)
Frederick L Brancati (fbrancati@jhmi.edu)
Charles W Heilig (cheilig@medicine.bsd.uchicago.edu)
Alan Shuldiner (Ashuldin@medicine.umaryland.edu)
Eric Boerwinkle (Eric.Boerwinkle@uth.tmc.edu)
Josef Coresh (coresh@jhu.edu)

Version: 4 Date: 15 December 2010

Author's response to reviews: see over
December 5, 2010

Josef Coresh, M.D. Ph.D. and Charles C. Hsu, M.D., Ph.D.
Welch Center for Prevention, Epidemiology and Clinical Research
Department of Epidemiology
Johns Hopkins University
2024 E. Monument St., Suite 2-600
Baltimore, MD 21205
(410) 955-0495 fax 410-955-0476, coresh@jhu.edu

Dr. Hans Zauner
Scientific Editor
Dr. Philippa Harris
Executive Editor-in-Training
BioMed Central Medical Genetics
Floor 6, 236 Gray’s Inn Road
London, WC1X 8HL
Tel: +44 (0) 20 3192 2013
e-mail: editorial@biomedcentral.com

MS: 1772243348310353
Version 3
Genetic variation of Glucose Transporter-1 (GLUT1) and nephropathy in 10,278 Caucasian and African-Americans: a case-control study in the Atherosclerosis Risk in Communities (ARIC) Study

Dear Drs. Zauner and Harris:

Thank you very much for your most recent comments regarding our manuscript titled “Genetic variation of Glucose Transporter-1 (GLUT1) and nephropathy in 10,278 Caucasian and African-Americans: a case-control study in the Atherosclerosis Risk in Communities (ARIC) Study”. We appreciate your patience.

We are delighted that you now consider the referees’ criticisms satisfactorily addressed and anticipate acceptance of our revised manuscript.

In response to your latest correspondence, we have formatted our abstract according to the journal guidelines. Additionally, upon our review, the revised manuscript does conform to the journal style.
We appreciate your patience, and the revised manuscript has been attached.

Thank you. Please do not hesitate to contact us if there are any further questions.

Sincerely,

Josef Coresh, M.D. Ph.D.  Charles Hsu, M.D. Ph.D.